Author's response to reviews

Title: The effect of Melatonin on bacterial translocation following ischemia/reperfusion injury in a rat model of superior mesenteric artery occlusion

Authors:

Murat Ozban (muratozban@yahoo.com)
Cagatay Aydin (cagatayaydin@yahoo.com)
Nural Cevahir (cevahir@pau.edu.tr)
Cigdem Yenisey (mozban@pau.edu.tr)
Onur Birsen (birsenonur@yahoo.com)
Gulistan Gumrukcu (caydin@pau.edu.tr)
Berrin Aydin (berrinaydin@hotmail.com)
Ibrahim Berber (iberber@hotmail.com)

Version: 5 Date: 12 December 2014

Author's response to reviews: see over
Please find revised additional request formatting:

1) We included a title page at the front of the manuscript;
2) We removed the tables as figure files and placed them at the end of the manuscript;
3) We provided separate figure legends at the end of the MS;
4) We included an Acknowledgement Section at the end of the MS.

Sincerely.

Murat Ozban.